Author's response to reviews

Title: Obesogenic environments: a systematic review of the association between the physical environment and adult weight status. The SPOTLIGHT project

Authors:

Joreintje D. Mackenbach (j.mackenbach@vumc.nl)
Harry Rutter (harry.rutter@lshtm.ac.uk)
Sofie Compernolle (sofie.compernolle@ugent.be)
Ketevan Glonti (ketevan.glonti@lshtm.ac.uk)
Jean-Michel Oppert (jean-michel.oppert@psl.aphp.fr)
Helene Charreire (helene.charreire@u-pec.fr)
Ilse de Bourdeaudhuij (ilse.debourdeaudhuij@ugent.be)
Johannes Brug (j.brug@vumc.nl)
Giel Nijpels (g.nijpels@vumc.nl)
Jeroen Lakerveld (je.lakerveld@vumc.nl)

Version: 2 Date: 17 October 2013

Author's response to reviews: see over
Concerning: Revision of submission

Dear dr. Pafitis,

With this letter I would to describe the changes made to the manuscript since yesterday:
• The abstract was formatted according to the BMC Public Health guidelines
• E-mailaddresses of all co-authors were added to the title page
• As the submission did not contain the correct manuscript, the correct manuscript was uploaded

I hope this information is sufficient.

Yours Sincerely,
On behalf of the co-authors,

Joreintje D. Mackenbach